Author's response to reviews

Title: Association of the MAOA Promoter uVNTR Polymorphism with Suicide Attempts in Patients with Major Depressive Disorder

Authors:

For-Wey Lung (forwey@seed.net.tw)
Dong-Sheng Tzeng (tzengds@seed.net.tw)
Mei-Feng Huang (lalalabon@gmail.com)
Ming-Been Lee (forweylung@gmail.com)

Version: 5 Date: 20 May 2011

Author's response to reviews:

May 20, 2011

The Editor,

BMC Medical Genetics

Dear Editors of BMC Medical Genetics,

Enclosed is our manuscript entitled “Association and Interaction of the MAOA Promoter uVNTR Polymorphism with Suicide Attempts in Patients with Major Depressive Disorder”. We have proofread the final version of the manuscript and made some minor changes to the typographical errors which we have found.

Sincerely,

For-Wey Lung,
Dong-Sheng Tzeng,
Mei-Feng Huang,
Ming-Been Lee

Correspondence to Professor Ming-Been Lee
No.2 Chung Cheng 1st Rd., Kaohsiung, Taiwan
Tel: 886 7 7490056; Fax: 886 7 7493767
E-mail: forweylung@gmail.com